Author's response to reviews

Title: Membranous nephropathy and lupus like syndrome after haematopoietic cell transplantation: a case report.

Authors:

Kostas Stylianou (kstylianu@gmail.com)
Stavros Stratakis (indian_steven@yahoo.gr)
Vasiliki Mavroeidi (v.mavroidi@yahoo.gr)
Ioannis Petrakis (petrakgia@gmail.com)
Dimitris Xydakis (dxydakis@gmail.com)
Eleftheria Vardaki (elevard@in.gr)
Spyros Stratigis (stratigiss@yahoo.com)
Kostas Perakis (perkost@gmail.com)
Theodora Katsarou (theokats@windowslive.com)
Peggy Kanellou (kanelloup@yahoo.gr)
Irene Xylouri (antonakisnikos@yahoo.gr)
Constantina Petraki (nelniko@otenet.gr)
Michael Alexandrakis (alexandm@med.uoc.gr)
Eugene Daphnis (eugenedaphnis@yahoo.gr)

Version: 5 Date: 24 June 2010

Author's response to reviews:

Major revisions

Patient information: please provide the age and ethnicity of the patient in the abstract section. Response: revision was made accordingly

References: page 4 cites a reference ' [0] ', please clarify the number of this reference.
Response: we are not able to find this citation in the text. Please if you keep seeing this reference, just delete it. There must not be such a reference in the text.

Minor revisions.

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods).
Response: revisions were made

Abstract: please remove the keywords section. Keywords have now been removed

Figure legend: please place this section after the references. The legend has
been placed after references

Manuscript format: please remove all the review tracking. Review tracking has been removed now.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Response: we can't find any typographical errors. A few abbreviations have been added in the "List of abbreviations" section